Author’s response to reviews

Title: Do Bacteremic Patients with End-Stage Renal Disease Have a Fever When Presenting to The Emergency Department? A Paired, Retrospective Cohort Study.

Authors:

Sarah Weatherall (sarahlweatherall@gmail.com)

Alison Chambers (Alison.Chambers@Lifespan.org)

Leonard Mermel (lmermel@lifespan.org)

Version: 3 Date: 26 Dec 2019

Author’s response to reviews:

Dear Editors,

We have made the requested change to Keywords and added - All authors have read and approved the final manuscript. We confirm that we have permission to include the one person mentioned in the Acknowledgements section.

We hope this meets with your approval.

Sincerely,

Leonard A. Mermel, DO, ScM, AM (Hon), FSHEA, FIDSA, FACP
Professor of Medicine, Warren Alpert Medical School of Brown University
Medical Director, Dept. of Epidemiology &amp; Infection Control, Rhode Island Hospital
Adjunct Clinical Professor, University of Rhode Island College of Pharmacy